Author’s response to reviews

Title: EmPhAsIS: Empowering Pharmacists in Asthma management through Interactive SMS. Study protocol for a pragmatic, pharmacy-based, cluster randomized controlled trial.

Authors:

Mary A De Vera (mary.devera@ubc.ca)
Mohsen Sadatsafavi (mohsen.sadatsafavi@ubc.ca)
Nicole W Tsao (nicole_t@mail.ubc.ca)
Larry D Lynd (larry.lynd@ubc.ca)
Richard Lester (rlester@mail.ubc.ca)
Louise Gastonguay (louisega@mail.ubc.ca)
Jessica Galo (jessica.galo@ubc.ca)
J. Mark FitzGerald (Mark.Fitzgerald@vch.ca)
Penelope Brasher (Penny.Brasher@vch.ca)
Carlo A Marra (cmarra@mun.ca)

Version: 2 Date: 15 August 2014

Author’s response to reviews:

See previous cover letter